Author's response to reviews

Title: Clinical outcomes associated with Staphylococcus aureus and Pseudomonas aeruginosa airway infections in adult cystic fibrosis patients

Authors:

Heather G Ahlgren (heather.ahlgren@gmail.com)
Andrea Benedetti (Andrea.Benedetti@mcgill.ca)
Jennifer S Landry (jennifer.landry@mcgill.ca)
Joanie Bernier (joanie_bernier1@hotmail.com)
Elias Matouk (elias.matouk@mcgill.ca)
Danuta Radzioch (danuta.radzioch@gmail.com)
Larry Lands (larry.lands@mcgill.ca)
Simon Rousseau (Simon.Rousseau@mcgill.ca)
Dao Nguyen (dao.nguyen@mcgill.ca)

Version: 2 Date: 17 March 2015

Author's response to reviews:

Dear Editor to the BMC Pulmonary Medicine,

As requested, we have made the following changes:
1. A conclusion heading was added.
2. A section and statement for Competing interests were added.
3. A list of abbreviations was already present and no changes were made.